\documentclass[kind=ENG,a4j,graduate,landscape,viewer,color,check]{edb-syl}
\def\�N�x{2011}
\def\�a�N�x{23}
\begin{document}

\Englishfalse
%%%% 216539:216596
\begin{edb:lecture}%
\EdbEID{216596}\EdbEOID{538053}%
\JabeeEfalse%
\JabeeRfalse%
\���false%
\����ID{216539:216596}%
\���ƊT�v���{166222}%
\����{�ۑ�T���@(D)}{Venture Business (D)}%
\�S������{}%
\�J�u�w��{}%
\���C���{�I��}%
\EdbColDef{edb:lecture.goal}{1}{E}{Objective of the lecture is to have experience and knowledge of collaborative works with private enterprises and to mature the works to venture businesses.}%
\EdbColDef{edb:lecture.goal}{1}{J}{��ƂƂ̋���������C�����ʂ����x���`���[�r�W�l�X�֓W�J�����o��$\cdot{}$�m����L���邱�ƁD}%
\EdbColDef{edb:lecture.unit}{1}{E}{2}%
\EdbColDef{edb:lecture.unit}{1}{J}{2}%
\EdbColDef{edb:lecture.target}{1}{E}{Aims of the lecture are doing collaborative works with private enterprises and establishing venture businesses through such works.}%
\EdbColDef{edb:lecture.target}{1}{J}{��ƂƂ̋��������̎��{��C�����ʂ����x���`���[�r�W�l�X�ւ̓W�J��ړI�Ƃ���D}%
\EdbColDef{edb:lecture.outline}{1}{E}{The lecture is aiming at founding a venture business company as well as establishing collaborative research works with private enterprises.\ It's worth to apply their accomplishment to an actual society beyond the range of a laboratory work.}%
\EdbColDef{edb:lecture.outline}{1}{J}{����̌������x���`���[�r�W�l�X�ɔ��W�������ꍇ�ɔF�肷����ƉȖڂł���D���������ł̌����Ɏ~�܂炸�C���̐��ʂ��Љ�ɖ₢������s���܂Ő��������߂����Ƃ͍����]���ɒl����Ƃ��������ɂ��D}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{Credits are ackowledged in case of doing collaborative works with enterprises and/or maturing them to venture businesses.\ Handouts,{\EdbZSp} reports,{\EdbZSp} and presentation materials,{\EdbZSp} etc.\ are evaluated to approve credits.}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{����̌�������ƂƋ�������������C�x���`���[�r�W�l�X�ɔ��W�������ꍇ�C�P�ʂ�F�肷��D�ł����킹�ɗp������������]���̑ΏۂƂ���D}%
\EdbColDef{edb:lecture.note}{1}{E}{A possible student should submit portfolios which are evidenced for his/her 90 hours' extensive work.\ Students are advised to ask details about this lecture to their supervisers.}%
\EdbColDef{edb:lecture.note}{1}{J}{90���ԕ��̊w�C�L�^(�|�[�g�t�H���I)���o����D�ڍׂɂ��ẮC�w�������ɑ��k�C�w���������ƁD}%
\end{edb:lecture}%

\Englishtrue
%%%% 216539:216596
\begin{edb:lecture}%
\EdbEID{216596}\EdbEOID{538053}%
\JabeeEfalse%
\JabeeRfalse%
\���false%
\����ID{216539:216596}%
\���ƊT�v���{166222}%
\����{Venture Business (D)}{�ۑ�T���@(D)}%
\�S������{}%
\�J�u�w��{}%
\���C���{selection}%
\EdbColDef{edb:lecture.goal}{1}{J}{��ƂƂ̋���������C�����ʂ����x���`���[�r�W�l�X�֓W�J�����o��$\cdot{}$�m����L���邱�ƁD}%
\EdbColDef{edb:lecture.goal}{1}{E}{Objective of the lecture is to have experience and knowledge of collaborative works with private enterprises and to mature the works to venture businesses.}%
\EdbColDef{edb:lecture.unit}{1}{J}{2}%
\EdbColDef{edb:lecture.unit}{1}{E}{2}%
\EdbColDef{edb:lecture.target}{1}{J}{��ƂƂ̋��������̎��{��C�����ʂ����x���`���[�r�W�l�X�ւ̓W�J��ړI�Ƃ���D}%
\EdbColDef{edb:lecture.target}{1}{E}{Aims of the lecture are doing collaborative works with private enterprises and establishing venture businesses through such works.}%
\EdbColDef{edb:lecture.outline}{1}{J}{����̌������x���`���[�r�W�l�X�ɔ��W�������ꍇ�ɔF�肷����ƉȖڂł���D���������ł̌����Ɏ~�܂炸�C���̐��ʂ��Љ�ɖ₢������s���܂Ő��������߂����Ƃ͍����]���ɒl����Ƃ��������ɂ��D}%
\EdbColDef{edb:lecture.outline}{1}{E}{The lecture is aiming at founding a venture business company as well as establishing collaborative research works with private enterprises.\ It's worth to apply their accomplishment to an actual society beyond the range of a laboratory work.}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{����̌�������ƂƋ�������������C�x���`���[�r�W�l�X�ɔ��W�������ꍇ�C�P�ʂ�F�肷��D�ł����킹�ɗp������������]���̑ΏۂƂ���D}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{Credits are ackowledged in case of doing collaborative works with enterprises and/or maturing them to venture businesses.\ Handouts,{\EdbZSp} reports,{\EdbZSp} and presentation materials,{\EdbZSp} etc.\ are evaluated to approve credits.}%
\EdbColDef{edb:lecture.note}{1}{J}{90���ԕ��̊w�C�L�^(�|�[�g�t�H���I)���o����D�ڍׂɂ��ẮC�w�������ɑ��k�C�w���������ƁD}%
\EdbColDef{edb:lecture.note}{1}{E}{A possible student should submit portfolios which are evidenced for his/her 90 hours' extensive work.\ Students are advised to ask details about this lecture to their supervisers.}%
\end{edb:lecture}%

\end{document}
\end{document}


